Author's response to reviews

Title: Radiation Retinopathy: Case Report and Review

Authors:

Abha Gupta (abha16@hotmail.com)
Felipe Dhawahir-Scala (felipe.dhawahir@btinternet.com)
Amy Smith (amysmith1@talk21.com)
Lorna Young (lorna.young@cmmc.nhs.uk)
Steve Charles (SalisburyHouseAE@aol.com)

Version: 6 Date: 4 April 2007

Author's response to reviews: see over
Dear Editor,

I have amended the manuscript as detailed in the email you sent me. Below are the changes made. Please note figure 1 is now amended also.
If there are any concerns please email me on abha16@hotmail.com

1. A title page has been included in the manuscript. It contains Title, Author list, Affiliations, E-mail addresses, corresponding first author (along with email address)
2. The authors and affiliations have been added in the style requested
3. The abstract has been included in the style requested
4. The manuscript headings are in keeping with the journals request, they are no longer underlined
5. The figure legends have been moved to after the references
6. The table is now after the figure legends
7. I have stated all authors have read the manuscript
8. I have moved the competing interests and author contributions to before the references
9. I have cited the references in the text using square brackets
10. please note figure 1 is now amended

Kind regards
Abha Gupta